Author’s response to reviews

Title: A population-based surveillance study on severe acute maternal morbidity (near-miss) and adverse perinatal outcomes in Campinas, Brazil: The Vigimoma Project.

Authors:

Eliana Amaral (elianaa@unicamp.br)  
João Paulo Souza (souzaj@who.int)  
Fernanda Surita (surita@unicamp.br)  
Adriana G Luz (adrigoluz@terra.com.br)  
Maria Helena Sousa (mhestat@cemicamp.org.br)  
José Guilherme Cecatti (cecatti@unicamp.br)  
Oona Campbell (Oona.Campbell@lshtm.ac.uk)

Version: 5 Date: 18 January 2011

Author’s response to reviews:

Dear Sirs,

I have found few typing errors which were corrected, including an extra comma at page 15 and a misplaced reference, with no number typed.

The new version is attached below.

Many thanks,

Eliana Amaral  
Corresponding author